Check for updates

OPEN ACCESS

EDITED BY Mohammed A. Al-Sharafi, University of Technology Malaysia, Malaysia

REVIEWED BY

Ahmad Samed Al-Adwan, Al-Ahliyya Amman University, Jordan Mostafa Al-Emran, British University in Dubai, United Arab Emirates Ra'ed Masa'deh, The University of Jordan, Jordan Ibrahim Arpaci, Bandirma Onyedi Eylül University, Türkiye

*CORRESPONDENCE Masoud Yazdanpanah ⊠ yazdanm@asnrukh.ac.ir

SPECIALTY SECTION This article was submitted to Educational Psychology, a section of the journal Frontiers in Psychology

RECEIVED 21 November 2022 ACCEPTED 26 January 2023 PUBLISHED 17 February 2023

CITATION

Zobeidi T, Homayoon SB, Yazdanpanah M, Komendantova N and Warner LA (2023) Employing the TAM in predicting the use of online learning during and beyond the COVID-19 pandemic. *Front. Psychol.* 14:1104653. doi: 10.3389/fpsyg.2023.1104653

COPYRIGHT

© 2023 Zobeidi, Homayoon, Yazdanpanah, Komendantova and Warner. This is an openaccess article distributed under the terms of the Creative Commons Attribution License (CC BY). The use, distribution or reproduction in other forums is permitted, provided the original author(s) and the copyright owner(s) are credited and that the original publication in this journal is cited, in accordance with accepted academic practice. No use, distribution or reproduction is permitted which does not comply with these terms.

Employing the TAM in predicting the use of online learning during and beyond the COVID-19 pandemic

Tahereh Zobeidi¹, Seyedeh Bahar Homayoon², Masoud Yazdanpanah^{1,2,3}*, Nadejda Komendantova¹ and Laura A. Warner³

¹Cooperation and Transformative Group, International Institute for Applied Systems Analysis (IIASA), Laxenburg, Austria, ²Department of Agricultural Extension and Education, Agricultural Sciences and Natural Resources University of Khuzestan, Mollasani, Iran, ³Department of Agricultural Education and Communication, Institute of Food & Agricultural Sciences, University of Florida, Gainesville, FL, United States

Online learning systems have become an applied solution for delivering educational content, especially in developing countries, since the start of the COVID-19 pandemic. The present study is designed to identify the factors influencing the behavioral intention of agricultural students at universities in Iran to use online learning systems in the future. This research uses an extended model in which the constructs of Internet self-efficacy, Internet anxiety, and output quality are integrated into the technology acceptance model (TAM). Data analysis was performed using the SmartPLS technique. The analyses showed the proposed model to be strong in terms of predicting the attitude to online learning and the intention variance. Our findings show attitude and perceived usefulness to have directly affected intention. Output quality and Internet self-efficacy indirectly affected attitude and intention. Research findings can help with the design of educational policies and programs to facilitate education and improve student academic performance.

KEYWORDS

internet anxiety, internet self-efficacy, psychology, agricultural education, TAM

1. Introduction

COVID-19 first appeared in December 2019 in Wuhan, China. Based on World Health Organization (WHO) statistics, the deadly disease was infecting millions and leaving thousands dead (World Health Organisation, 2021; Hallaj et al., 2022; Moradhaseli et al., 2022). As a response to the COVID-19 crisis, governments around the world have been implementing public policies such as social distancing, isolation, and quarantine (Anderson et al., 2020; Amghani et al., 2022). Many Asian and European countries have shut down large communities, including educational institutions, to combat this "invisible" disease (Sahu, 2020). In these circumstances, universities have turned to advances in information technology as a possible means of overcoming the difficulties being encountered (Raza et al., 2021). COVID-19 has been unprecedented in changing the way students are taught around the world within a short period of time (Chung et al., 2020). Many universities have started to use online learning as a way of ensuring continuity of education (Chung et al., 2020), and have moved rapidly from face-to-face courses and programs to delivery of educational content online (Gewin, 2020). Iran is one of the countries most affected by the outbreak of COVID-19. To prevent the transmission of the virus, the Iranian government closed educational institutions on 5 March 2020 (Taghizadeh et al., 2021).

Online learning is an established, effective, and influential learning medium in many educational organizations and institutions (Lee et al., 2020). Online learning, an advanced e-learning method, includes training, coaching, information, and any learning content that is presented digitally or electronically (Fallon and Brown, 2002). Online learning programs, such as chat meetings, posts, and emails, facilitate communication between teachers and students in a variety of ways (Romanov and Nevgi, 2006). In Iran, the University of Tehran, with the establishment of the Electronic Education Center in 2002, is the first university that started studying and planning online education, but the first collegiate virtual course was held in Shiraz University in 2003, and subsequently in the universities of science and technology, Khajeh Nasir al-Din Toosi University of Technology, and Amir Kabir. Most of the large public universities in Iran have established e-learning centers so that the quantitative growth of students in e-courses has been increasing from 235 in 2013 to 19,000 in 2013 (Sharifi et al., 2019). It is argued that COVID-19 accelerates the digital transition in the world, which provides solutions regardless of the physical world (Arpaci et al., 2022). In Iran, the Covid-19 pandemic led to the flourishing of some capabilities in the country, including the spread and prosperity of virtual education throughout the country, so that virtual education in Iran has entered a new phase and more attention has been paid to virtual education (Yaghoubi Farani et al., 2019). On the other hand, those in charge have also become more aware of the importance of distance education and e-learning education (Ghafourifard, 2020). Online learning provides learning opportunities anytime and anywhere (Zamani-Miandashti and Ataei, 2015; Joosten and Cusatis, 2020); it focuses on learners' needs and preferences and also activates critical thinking on the part of students. This type of learning environment also increases the efficiency and success of learning facilities by improving communication between students and educators (Idris and Osman, 2015). These benefits, together with the increasing demands being placed on higher education establishments regarding the admission of new students and how their needs can be met, are the justification for online education (Kim et al., 2005). The flexible use of online media does, however, require students to have skills such as knowledge of technology use, time and organization management, and interaction using online technologies (Joosten and Cusatis, 2020). Educational institutions are thus investing in information systems to offer increased access to education and to improve students' self-efficacy, knowledge production, cost-effectiveness, together with learner flexibility and engagement (Sinclair et al., 2016). However, if the e-learning system is not used properly, it is difficult to derive its full benefits (Almaiah and Al-Khasawneh, 2020). Such could also lead to the failure of the education system and reduce universities' return on investment (Naveed et al., 2017). In fact, the success of the e-learning system relies on the intention and readiness of students to use it (Shawai and Almaiah, 2018; Almaiah and Alismaiel; Al-Adwan et al., 2021). Research into online learning is still in its infancy (Tarhini et al., 2017; Almaiah and Alamri, 2018). Participation in e-learning needs more in-depth study so that the requirements of university students can be better understood: this will ultimately lead to a more successful e-learning system (Alksasbeh et al., 2019).

Despite the importance of examining students' intention to use online learning systems, there are few literature reviews on the intention to use – and the actual use of – online learning (Mouakket and Bettayeb, 2016; Al-Adwan et al., 2021). As, to our knowledge, there is no research that examines the intention of Iranian students to use online learning and its predictors, the present study aims to address these two issues critical gaps in knowledge.

Many theoretical frameworks have been applied to cognitive aspects of the use of technology in education (Teo et al., 2018). Theoretical models, such as the task-technology fit model (Venkatesh et al., 2003), social cognitive model (Bandura, 1986), the theory of planned behavior (Ajzen and Fishbein, 1975), and the technology acceptance model (TAM) (Venkatesh and Davis, 2000) have been applied to study students' attitudes, perceptions, and behaviors regarding online learning. Of these, TAM is reported to be the best broadly functional model in the arena of social sciences (Teo et al., 2018). Although TAM has been lately disapproved for being an obsolete model, the bibliometric analysis study of (Al-Emran and Granić, 2021) specified that the number of research's on TAM and its applications are on the rise, providing indication the model is still valid across several applications and fields.

Several years after the advent of TAM, this model was expanded, with various external variables being added to it (Rahmawati, 2019). The expansion was critical to TAM, as TAM did not then pay enough attention to individual characteristics (Agarwal and Prasad, 1999; McMaster and Wastell, 2005; Cheng, 2011; Khayun and Ractham, 2011). Scientists have conducted studies with the extended TAM model, using differing variables in different countries and different fields of e-learning and especially during and beyond Covid-19 (Sukendro et al., 2020; Al-Adwan et al., 2021, 2022; Aboagye et al., 2021; Mailizar et al., 2021; Al-Nuaimi et al., 2022; Han and Sa, 2022). Al-Adwan et al. (2021, 2022) focused on quality factors of e-learning systems, and self-directed learning in the extended TAM. Han and Sa (2022) used educational satisfaction instead of attitude to explain acceptance intention of online education. Sukendro et al., (2020) also added facilating conditions to the TAM. In addition, facilitating conditions and major barriers are effective in the adoption of online learning (Sukendro et al., 2020). One of these conditions is the availability of adequate internet speed, especially in developing countries. Iran's situation in terms of internet speed has been inadequate. Among 211 countries in the world, Iran ranks 117th in terms of internet speed (Lahordi, 2021). Therefore, concern about the state of the Internet is considered one of the most important factors that can affect the acceptance of online education in Iran. To our best knowledge, there is not any study in Iran which considered the perceived anxiety and internet self-efficacy along with the dimensions of the outputs quality. The novelty of this study is considering the dimension of internet anxiety and internet self-efficacy considering low Internet speed in Iran. Precisely, this study's contribution is two-fold. First, the study goals to discover agricultural students' intention to adopt onlinelearning and the conditional factors including internet anexity and internet self-efficacy that affect their intention during and beyond the COVID-19 disaster. Second, the study pursues to inspect online learning performances based on empirical data from a developing country, Iran.

The contribution of this study extends to university policy makers and educators who can gain deeper insight and understanding of student acceptance of online learning technology beyond Covid-19, leading to better e-learning policy development.

This study uses the TAM, extended by specific constructs, to predict the intention to use online learning beyond Covid-19. To achieve this general goal, we pursued the following specific objectives:

- Use of TAM in predicting students' intention to use online learning.
- Use of the expanded TAM to improve the explanatory power of the intention to use online learning.

The paper is organized as follows. Section 2 outlines the literature review with the main components of the theoretical framework and its expanded version. In this section, we explain TAM as the underlying framework to examine psychological factors affecting students' adoption of online learning. Specifically, we corporate the TAM with the constructs of output quality, perceived internet anxiety, and internet selfefficacy, which has been explained in this section. Section 3 discusses the methodology including case study features, constructs measurements, and data analysis. Section 4 presents the results, section 5 discusses the finding, and section 6 discusses the theoretical and practical implications for the development use of online education systems in higher education. Finally, the conclusion and limitations are discussed in section 7.

2. Literature review

2.1. Technology acceptance model (TAM)

TAM, which was introduced by Fred D. Davis in 1986, is considered to be the greatest influential theory describing the acceptance of information systems (Lee et al., 2003). A well-known theory on the use and prediction of information technology acceptance, its capability has been validated by several studies in the domain of e-learning (Lee et al., 2011; Padilla-Meléndez et al., 2013; Lee et al., 2014; Tarhini et al., 2014; Wu and Zhang, 2014; Al-Gahtani, 2016; Rahman et al., 2016). The model is rooted in the Theory of Reasoned Action and Theory of Planned Behavior which were proposed by Ajzen and Fishbein (1980), Sarcheshmeh et al. (2018), and Ajzen (1985), respectively, under the theories of Social Psychology (Al-Fraihat et al., 2020). TAM has been formed as an analytical, and influential model for amplification and forecasting performance in decision-making as well as acceptance of using a particular technology (Savari et al., 2021). Based on the TAM, the current and future use of online learning systems is determined by two variables, perceived usefulness (PU) and perceived ease of use (PEOU) (Davis, 1993). In the TAM, PU refers to a person's belief that the use of technology improves their work performance (Davis, 1989), and PEOU is defined as the degree to which one believes technology is easy to use and free from effort (Davis, 1989). Intention is defined as a cognitive process, based on people's readiness to perform a target behavior (Abbasi et al., 2011).

The literature on information systems indicates that higher PEOU leads to higher PU. Indeed, when students perceive using the Internet is simple—they think it is very useful (Ebrahimi et al., 2018). This positive and significant relationship between the two variables is also supported by research in the field of e-learning (Lee et al., 2009; Faqih, 2013; Ibili et al., 2019; Demoulin and Coussement, 2020).

According to TAM, PU and PEOU affect people's attitudes toward using technology. These relationships are supported by different studies in mobile technology (Briz-Ponce et al., 2017), learning management systems (Revythi and Tselios, 2019), cloud services (Huang, 2016), and e-learning (Al-Rahmi et al., 2019; Salloum et al., 2019). Arpaci (2017) also found PU and PEOU affect attitudes toward using distance education tools and systems.

Attitude refers to an individual's assessment (favorable/ unfavorable and or positive/negative feelings) regarding the performance of a given behavior (Shahangian et al., 2021; Valizadeh et al., 2021; Ataei et al., 2021a,b). According to research (Sukendro et al., 2020), attitudes toward the use of online education during Covid-19 could affect a person's behavioral intentions to use online education.

Buabeng-Andoh (2021) examined the behavior of students in developing countries and investigated their behavioral intention regarding the use of mobile learning, showing that attitude has a great impact on behavioral intentions: if students have a positive attitude toward technology, they are more likely to use it. In particular, research conducted in developing countries, such as Chang et al. (2017) in Azerbaijan, Hussein (2017) in Malaysia, and Hanif et al. (2018) in Pakistan, found attitude to be a significant predictor of the intention to use learning systems. Research has shown that PU also has a direct and positive effect not only on attitude but also on behavioral intentions. Studies have found PU to have a positive effect of on intention toward using electronical and multimedia online learning systems (Al-Rahmi et al., 2019; Hariguna, 2019; Salloum et al., 2019; Teo et al., 2019). The study results of Al-Adwan (2020) showed students' behavioral intention to adopt massive open online courses is positively affected by the PEOU and PU. Al-Emran et al. (2020) also found that PEOU and PU have significant effects on students' behavioral intention to use smartwatches for educational purposes.

2.2. Expanding the technology acceptance model

TAM focuses primarily on technology-related features but ignores other key non-technological factors, such as individual traits, that play a critical role in the use of technology. In the current research, therefore, steps have been taken to expand the TAM model using two constructs, "Internet self-efficacy" and "Internet anxiety," which are based on individual characteristics, plus another construct, namely, "output quality," which also plays a key role in using online learning systems.

Individual characteristics are significantly diverse. A vital individual variable in terms of technology use is self-efficacy (Mahdavian et al., 2016). The first individual construct added to TAM was Internet self-efficacy (Cheng, 2011; Zhao et al., 2011). Self-efficacy is defined as a person's belief they can successfully carry out the necessary actions to produce the desired results in a certain context (Yazdanpanah et al., 2020; Karataş et al., 2022; Savari et al., 2022; Tajeri Moghadam et al., 2022) and Internet self-efficacy refers to a person's judgment regarding their ability to successfully carry out actions on the Internet (Compeau and Higgins, 1995; Eastin and LaRose, 2000; Yuen and Ma, 2008). Internet self-efficacy is an important predictor in TAM (Venkatesh and Bala, 2008; Chung et al., 2010; Chang and Im, 2014; Tsai et al., 2016). Studies (Park, 2009; Isaac et al., 2017) have demonstrated significant effects of Internet self-efficacy on PU and PEOU.

The second individual construct added to the model is Internet anxiety; anxiety regarding technology is linked to the person's inner self-control (Hernández-García and Iglesias-Pradas, 2012). This variable is also related to the undesirable emotions a person can experience when they work with technology (Sánchez-Prieto et al., 2017). Internet anxiety itself comes from the construct of computer anxiety. Computer anxiety in the education field is defined as restlessness, apprehensiveness, or fear of present or future computer use (Igbaria and Parasuraman, 1989) in education (Sánchez-Prieto et al., 2017). Alenezi and Karim (2010) argue that computer anxiety plays a major role in the acceptance of e-learning in higher education organizations. Studies have indicated that anxiety is related to avoidance of e-learning systems or technologies or to using them infrequently (Chen and Tseng, 2012; Park et al., 2012; Purnomo and Lee, 2013). More anxious. students are less likely to use e-learning systems (Al-Alak and Alnawas, 2011). In particular, studies have supported the negative effect of computer anxiety on PEOU (Karaali et al., 2011; Chen and Tseng, 2012; Park et al., 2012; Ali et al., 2013; Calisir et al., 2014; Al-Gahtani, 2016). Research has also confirmed that anxiety has a negative effect on PU (Park et al., 2012; Purnomo and Lee, 2013). The more anxious students are, the less useful they perceive a computer to be.

The third construct added to the TAM is output quality. Output quality refers to the degree to which a person thinks or believes a system works satisfactorily and is defined as the level of positive effects of a person's use of an information system (Venkatesh and Davis, 2000). Several studies indicated that output quality influences PU and PEOU (Davis et al., 1992; Venkatesh and Davis, 2000; Lee et al., 2018). Meanwhile Ji et al. (2019) argue that the quality of outputs is essential for users to continue using online learning programs. Figure 1 shows the research extended TAM employed by the present study which was drawn based on a vast literature review.

3. Methodology

3.1. Participants and sample

The present research was performed using a cross-sectional study. The statistical population of the study consisted of agricultural students in Iran. According to the Institute for Research and Planning in Higher Education of Iran, 115,261 students are studying different fields of agriculture and at different academic levels in the 2020–2021 academic year. Data collection was done in the summer of 2020 after holding of the second virtual education course since the COVID-19 outbreak. An online survey was designed, and a convenience sampling method was used to distribute it online among agricultural students at Iranian universities (N = 115,261). Various tools including information sharing platforms, such as WhatsApp and Telegram, forums, and other sites

were used to distribute the questionnaire. Finally, 480 completed questionnaires were collected; the response rate was 51%. Based on Cochran's equation, 383 samples are enough for the research target population, however, in this study, more samples were selected to ensure the findings. Descriptive statistics on students' individual characteristics showed that the average age of students was about 24.5 years (SD = 6.46). The minimum age was 18 and the maximum was 54. The mean family size was 4.71 (SD = 1.66). Of 480 respondents, 69.6% were women and 30.4% were men. Of all respondents, 365 were undergraduate students (76.5%), 13.6% were master's students, and 9.9% were doctoral students. Of the respondents, 84.1% had no experience of using virtual training prior to the COVID-19 outbreak, while 15.9% had participated in online classes.

3.2. Questionnaire development

In this study, a structured two-part questionnaire was applied to examine the research framework. The first section consisted of questions related to the demographic information of the participants. The second measured the variables of the research framework. Items related to all constructs were quantified using 5-point Likert scales. The response choices for all items except Internet self-efficacy ranged from "strongly disagree" (1) to "strongly agree" (5). For the Internet self-efficacy items, students were requested to specify how confident they were in their ability to use the Internet for each of the alternatives and to indicate their responses on a scale from "very low" (1) to "very high" (5).

To ensure the validity of the questionnaire, a group of experts from different agricultural and environment education fields edited and approved the questions. Cronbach's alpha reliability coefficients obtained through a pilot study involving 30 students indicated reliability from very good to excellent, generally 0.81–0.912 for all scales.

The questionnaire consisted of 30 total items that addressed the factors included in the research model; some items with low item-toitem correlation were omitted in the statistical analysis and 25 were used



TABLE 1 Items of the extended TAM model and their sources.

Items	Source			
Internet anxiety				
I get confused when working with the internet	Calisir et al. (2014) and Sánchez-Prieto et al. (2017)			
I get anxious when I think about taking an online course				
I get nervous when I am asked to participate in online discussions				
I'm worried about enrolling in online courses				
I avoid working on the internet				
I get anxious when I need to use online resources				
I get nervous about getting lost in virtual space				
Intention				
I plan to take virtual classroom systems courses in the future if held at the university	Tao et al. (2019), Park (2009) and Davis (1989)			
I intend to be an active user in an online-learning system				
I recommend that some courses be held virtually after COVID-19				
Internet self-efficacy				
How confident are you about your ability in each of the following?	Kuo (2010) and Kuo et al. (2014)			
Troubleshooting Internet hardware				
Learning advanced skills in an online course				
Creating an online chat group if needed				
Understanding Internet hardware terms/words				
Attitude				
I think it was a good idea to attend the online classes	Sivo et al. (2018), Calisir et al. (2014), Park (2009) and Davis (1989)			
Output quality				
The scientific quality of the online classes was the same as the face-to-face classes I had	Eom et al. (2006)			
I learned as much from online classes as I did from the face-to-face version of the course				
The quality of the learning experience in online classes is better than in-class classes				
PEOU (PEOU)				
It was easy for me to learn how to use online classes	Sivo et al. (2018), Calisir et al. (2014), Tao et al. (2019), Park (2009)			
I think the process of using online classroom systems is clear and understandable	Davis (1989) and Pal and Vanijja (2020)			
Acquiring skills using online classroom systems is easy for me				
PU (PU)				
Online classes provided me with a valuable learning experience	Davis (1989) and Pal and Vanijja (2020)			
Online classes minimize inequalities in education				
Assessing the success of online classes is a somewhat objective exercise				

in the final analysis. Details of the scale of the items and the literature source for each construct are provided in Table 1.

3.3. Data analysis

We used structural equation modeling (SEM) to explore the associations among constructs. SEM, a path analysis with latent variables, is a method that is commonly used in the behavioral and social sciences to validate and test hypothetical or theoretical models (Yazdanpanah et al., 2021; Zobeidi et al., 2022). SEM technique contained of factor analysis and multiple regression analysis and was functional to assess the structural relationship between the measured and latent variables. SEM is a version of regression that includes a "measurement model" for some variables in general analysis. These variables, rather than being represented by a single-item, are represented by multiple items that are "weighted" in a manner similar to factor analysis. The benefit of SEM is that these thoughts are usually more consistent than single-item indexes (Zobeidi et al., 2021).

SEM is generally divided into covariance-based modeling (CB-SEM) and partial least squares (PLS-SEM). To select the most appropriate analysis technique, Kolmogorov–Smirnov and Shapiro–Wilk normality tests were used to evaluate the normality of the sample. The results indicate that the hypothesis of normal data distribution at the level of 1% is rejected. Moreover, kurtosis and skewness coefficients showed that the range of kurtosis coefficients was from -1.509 to -0.44 and the range of skewness coefficients from -0.093 to 0.567. To evaluate the model, because of the lack of sample distribution normality, the PLS-SEM technique was used with SmartPLS software version 3.2. We bootstrapped 2,000 samples to measure the significance of the path coefficients.

TABLE 2	Description	of items.
---------	-------------	-----------

	Mean	SD	Factor loading	t value	VIF
Internet anxiety	2.71	1.0315			
Anx1	2.49	1.173	0.751	28.859	2.064
Anx2	2.99	1.380	0.852	60.619	3.060
Anx3	2.94	1.309	0.840	48.339	2.730
Anx4	2.92	1.327	0.882	79.420	3.548
Anx5	2.24	1.174	0.717	28.511	1.795
Anx6	2.53	1.282	0.744	27.239	2.091
Anx7	2.94	1.366	0.776	30.470	2.091
Itention	2.67	1.339			
In1	2.58	1.421	0.945	160.071	4.114
In2	2.81	1.387	0.913	92.531	3.057
In3	2.62	1.542	0.899	73.964	2.773
Internet efficacy	2.77	0.923			
IE1	2.36	1.106	0.797	39.576	2.316
IE2	2.57	1.147	0.745	24.876	1.981
IE3	3.21	1.212	0.874	85.855	1.835
IE4	2.94	1.153	0.760	30.692	1.557
Attitude					
Att1	2.80	1.501	1	-	1.000
Output quality	2.24	1.218			
Q1	2.26	1.309	0.913	82.361	3.433
Q2	2.28	1.326	0.932	114.048	3.785
Q3	2.23	1.311	0.899	71.051	2.458
Perceived ease	2.93	1.253			
to use					
PEOU1	2.97	1.346	0.929	110.993	3.442
PEOU2	2.81	1.360	0.929	110.685	3.408
PEOU3	3.02	1.318	0.926	104.844	3.531
Perceived	2.61	1.187			
usefulness					
PU1	2.68	1.373	0.888	78.184	2.205
PU2	2.56	1.360	0.902	76.991	2.586
PU3	2.63	1.241	0.891	68.919	2.501

4. Results

4.1. Measurement model

To test the extended TAM, the PLS-SEM approach was used due to the non-normal distribution of data. PLS-SEM was performed using SmartPLS software, version 3.2. First, factor analysis was used to scan the reliability of items. As shown in Table 2, the factor loading of all items surpassed the acceptable threshold of 0.7 (Nunnally, 1978). The items thus had good reliability.

The convergent and discriminant validity of the model were also assessed. Composite reliability (CR) and average variance extracted (AVE) indices were applied to confirm convergent validity. Table 3 shows that all values were higher than 0.7 for CR and 0.5 for AVE, thus indicating convergence validity (Fornell and Larcker, 1981).

The Fornell–Larcker criterion was used to assess discriminant validity. Based on this criterion, discriminant validity confirmed when the variance of the variables of a model was higher than the shared variance of each variable with its items (Fornell and Bookstein, 1982). To calculate the discriminant validity, the square root of each construct was compared with the correlation between the variables. As Table 3 shows, all variables meet the Fornell–Larcker criterion.

In addition to the Fornell-Larcker criterion, HTMT method was used to check the discriminant validity. There are two methods on HTMT ratio and HTMT inference which used to evaluate discriminant validity with HTMT. The HTMT ratio method is to calculate the HTMT ratio through the application of the PLS algorithm from the original sample data. In this method, to approve the discriminant validity, the indices obtained from the HTMT ratio should be less than 0.85 or 9.0 (Hair et al., 2019). The second method is the HTMT inference method, which is obtained by analyzing the confidence intervals obtained from HTMT after the bootstrap process. To endorse the discriminant validity using HTMT inference, the confidence intervals obtained should not contain a value of 1. As Table 4 shows, HTMT scores in all relationships were less than 0.9. Also, HTMT inference did not show a value of 1 within the confidence interval of any of the relationships. Therefore, the results confirmed the existence of discriminant validity.

Furthermore, it is vital to inspect the presence of multicollinearity at this phase. Multicollinearity between independent constructs can significantly distort the interpretation of the consequences. Therefore, VIF "Variance inflation factor" valuation is directed to regulate the presence of multicollinearity. To designate the nonappearance of multicollinearity, the metric of VIF for each independent variable should be <5, however ideally, the VIF values should be close to 3 and lower (Hair et al., 2019). As Table 4 shows, the absence of multicollinearity is confirmed, as the VIF for all variables are far less than the cut-off threshold of 5.

4.2. Structural model

After confirming the reliability and validity of the measurement model, the structural model of extended TAM was run. The structural model estimates the percentage of variance predicted by the model for endogenous constructs using the coefficient of determination (R^2); it also constructs cross-validated redundancy (Q^2), and the magnitude of the direct and indirect effects of the constructs using path coefficient and f square size, and their significance.

The Q² criterion was used to confirm the predictive relationship of the model through the Stone-Geisser test. According to Henseler et al. (2009), values of 0.02, 0.15, and 0.35 can be interpreted as low, medium, and strong predictive power, respectively. Chin (1998) points out that R^2 above the threshold of 0.19, 0.33, and 0.67 are weak, moderate, and substantial, respectively.

Based on R^2 and Q^2 shown in Tables 5, 6, the extended TAM is a strong model for predicting intention. The proposed model can predict 74% of the variance in students' intention to use online learning. This model also predicts about 62% of attitude variance through the effect of PU and PEOU. Moreover, 70 and 60% variance,

TABLE 3 Discriminant analysis and convergent validity.

	Output quality	Internet anxiety	Internet-efficacy	PU	PEOU	Attitude	Intention
Output quality	0.917						
Internet anxiety	-0.415	0.797					
Internet self-efficacy	0.406	-0.599	0.796				
PU	0.794	-0.402	0.456	0.894			
PEOU	0.645	-0.605	0.600	0.706	0.928		
Attitude	0.703	-0.404	0.458	0.755	0.687	1	
Intention	0.749	-0.439	0.465	0.783	0.732	0.824	0.919
Cronbach's alpha	0.905	0.905	0.812	0.874	0.919	1	0.908
Composite	0.94	0.924	0.873	0.922	0.949	1	0.942
reliability							
Average variance	0.84	0.635	0.633	0.799	0.861	1	0.845
extracted (AVE)							

Bolded elements indicate the square root of AVE.

TABLE 4 Results of the HTMT.

	HTMT ratio						
	Output quality	Internet anxiety	Internet- efficacy	Perceived usefulness	Perceived ease to use	Attitude	Intention
Output quality							
Internet anxiety	0.44						
Internet self- efficacy	0.437	0.676					
Perceived usefulness	0.89	0.428	0.494				
Perceived ease to use	0.705	0.651	0.659	0.786			
Attitude	0.738	0.402	0.462	0.806	0.716		
Intention	0.825	0.459	0.493	0.877	0.799	0.865	
				HTMT inference (2.5–97	.5%)		
Output quality							
Internet anxiety	0.348-0.527						
Internet self- efficacy	0.338-0.526	0.605-0.740					
Perceived usefulness	0.851-0.919	0.332-0.512	0.403-0.576				
Perceived ease to use	0.641-0.761	0.579-0.718	0.579-0.727	0.719-0.840			
Attitude	0.685-0.789	0.315-0.485	0.383-0.542	0.755-0.848	0.660-0.774		
Intention	0.775-0.862	0.369-0.539	0.408-0.578	0.835-0.915	0.737-0.849	0.821-0.904	

respectively, of PU and PEOU changes, are predicted by the extended TAM. Therefore, the research model is a strong model in terms of predicting intention.

The Cohen's f2 method was used to calculate the size of the constructs' effects. We examined the size of the proposed relationships between model constructs. Table 5 show the value of the path coefficients of the assumed relationships between the constructs and a summary of the hypotheses tested. The output quality variable was able to affect both the PU constructs ($\beta = 0.0.582$, $f^2 = 0.657$, t = 15.205) and the PEOU constructs ($\beta = 0.425$, $f^2 = 0.354$, t = 11.006) at the significant level of

p < 0.0001. The effect of output quality on PU and PEOU was thus found to be strong.

Internet anxiety also affected PU ($\beta = 0.087$, $f^2 = 0.014$, t = 2.481, p < 0.05) and PEOU ($\beta = -0.269$, $f^2 = 0.109$, t = 6.711, p < 0.0001), weakly and moderately, respectively. The effect of Internet anxiety on PU was positive, while on PEOU, it was negative.

The findings also showed that Internet self-efficacy affected PEOU ($\beta = 0.266$, $f^2 = 0.108$, t = 6.7, p < 0.0001), while having no effect on PU. PEOU, in turn, affected PU ($\beta = 0.344$, $f^2 = 0.159$, t = 7.209, p < 0.0001).

TABLE 5 Structural path analysis result.

Direct effect	β	Std. β	SD	T Statistics	f²	p values	Results
Output quality ->PU	0.568	0.582	0.039	15.205	0.657	0.000	Supported
Output quality -> PEOU	0.424	0.425	0.036	11.006	0.354	0.000	Supported
Internet anxiety - > PU	0.080	0.087	0.036	2.481	0.014	0.013	Supported
Internet anxiety - > PEOU	-0.263	-0.269	0.038	6.711	0.109	0.000	Supported
Internet self-efficacy ->PU	0.041	0.065	0.035	1.923	0.008	0.055	Rejected
Internet self-efficacy - > PEOU	0.277	0.266	0.038	6.7	0.108	0.000	Supported
PU -> Attitude	0.538	0.538	0.049	11.369	0.378	0.000	Supported
PU -> Intention		0.375		7.552	0.233	0.000	Supported
PEOU ->PU	0.368	0.344	0.050	7.209	0.159	0.000	Supported
Perceived ease of use -> Attitude	0.303	0.308	0.049	6.467	0.124	0.000	Supported
Attitude ->Intention	0.620	0.541	0.047	10.778	0.485	0.000	Supported
Indirect effects							
Output quality -> PU		0.146		5.829		0.000	Supported
Output quality -> Attitude	0.519	0.522	0.031	17.182		0.000	Supported
Output quality -> Intention	0.454	0.556	0.029	20.527		0.000	Supported
Internet anxiety - > PU	-0.097	-0.093	0.020	4.8		0.000	Supported
Internet anxiety - > Attitude	-0.089	-0.086	0.029	2.97		0.003	Supported
Internet anxiety -> Intention	-0.137	-0.049	0.028	1.713		0.087	Rejected
Internet efficacy -> PU	0.102	0.092	0.019	4.985		0.000	Supported
Internet self-efficacy - > Attitude	0.161	0.166	0.026	6.34		0.000	Supported
Internet self-efficacy -> Intention	0.186	0.149	0.026	5.556		0.000	Supported
PU -> Intention		0.291		7.836		0.000	Supported
PEOU ->Attitude	0.198	0.185	0.031	6.457		0.000	Supported
Perceived ease of use - > Intention	0.311	0.396	0.034	9.904		0.000	Supported
Total effects							
Output quality ->PU	0.724	0.728	0.028	27.618		0.000	Supported
Output quality -> PEOU	0.424	0.425	0.036	11.006		0.000	Supported
Output quality -> Attitude	0.519	0.522	0.031	17.182		0.000	Supported
Output quality ->Intention	0.454	0.556	0.029	20.527		0.000	Supported
Internet anxiety ->PU	-0.018	-0.006	0.038	0.157		0.875	Rejected
Internet anxiety -> PEOU	-0.263	-0.269	0.038	6.711		0.000	Supported
Internet anxiety -> Attitude	-0.089	-0.086	0.029	2.97		0.003	Supported
Internet anxiety - > Intention	-0.137	-0.049	0.028	1.713		0.087	Rejected
Internet self-efficacy - > PU	-0.018	0.157	0.038	4.491		0.000	Supported
Internet self-efficacy -> PEOU	0.277	0.266	0.038	6.7		0.000	Supported
Internet self-efficacy - > Attitude	0.161	0.166	0.026	6.34		0.000	Supported
Internet self-efficacy - > Intention	0.186	0.149	0.026	5.556		0.000	Supported
PU - > Attitude	0.538	0.538	0.049	11.369		0.000	Supported
PU ->Intention	0.335	0.666	0.045	19.938		0.000	Supported
Perceived ease to use ->PU	0.368	0.344	0.050	7.209		0.000	Supported
Perceived ease to use - > Attitude	0.501	0.493	0.044	11.403		0.000	Supported
Perceived ease to use -> Intention	0.311	0.396	0.048	9.904		0.000	Supported
Attitude ->Intention	0.620	0.541	0.047	10.778		0.000	Supported

PU ($\beta = 0.537$, $f^2 = 0.378$, t = 11.369, p < 0.0001) and PEOU ($\beta = 0.308$, $f^2 = 0.124$, t = 6.467, p < 0.0001), respectively, can strongly and moderately influence attitude toward participating in online learning

programs. Attitude ($\beta = 0.541$, f' = 0.485, t = 10.778, p < 0.0001) and PU ($\beta = 0.375$, f' = 0.233, t = 7.552, p < 0.0001) can predict the intention. Table 5 shows the direct, indirect, and total effect.

TABLE 6 Cross-validated redundancy of construct.

	Q ²
PU	0.527
PEOU	0.485
Attitude	0.60
Intention	0.59

5. Discussion

This study confirms the application of TAM as a strong theoretical framework for understanding predictors of the use of online learning systems. This study also takes a critical step in further developing the TAM with the three interpretive constructs of output quality, Internet anxiety, and Internet self-efficacy.

The findings confirm that these three constructs can influence the attitude and willingness to use online education systems through perceived usefulness and perceived ease. The proposed research model will predict 62 and 74 percent of attitude and intention variance changes, respectively.

The results showed that the behavioral intention to use online learning in the future is directly explained by people's attitudes toward using online learning and PU. Attitude was the strongest predictor of behavioral intention. Attitude is the reflection of people's feelings and appraisals relating to a concept or subject. Here, attitude refers to one's feelings about online learning systems. Students had greater intention to use online learning systems if they thought that doing so was a good idea and they felt positive about it.

Along with the research findings, several studies confirm the significant effect of attitude on the intention to use online learning (Lee et al., 2013; Calisir et al., 2014; Chang et al., 2017; Hussein, 2017; Hanif et al., 2018; Al-Rahmi et al., 2019; Buabeng-Andoh, 2021).

One of the factors that most affect attitude is PU: the results show the PU to strongly affect attitude. Phillips et al. (1994) show that PU reflects the psychological potential of end-users to consider using new technology as beneficial for personal and organizational well-being. In other words, when students find the use of online learning to be useful and beneficial, they develop a positive attitude toward using that system. Research confirms the significance and positive effect of PU on attitude (Huang, 2016; Briz-Ponce et al., 2017; Al-Rahmi et al., 2019; Revythi and Tselios, 2019; Salloum et al., 2019).

PEOU also has a positive effect on attitude, but this effect is not the same as that of PU— the effect size of PEOU on attitude was, in fact, moderate. This relationship has been confirmed in various studies (Huang, 2016; Briz-Ponce et al., 2017; Al-Rahmi et al., 2019; Revythi and Tselios, 2019; Salloum et al., 2019). Thus, PEOU and PU are shown to have directly influenced people's attitudes toward using online learning systems. PEOU also indirectly affects attitude and intention. As PEOU measures the user-friendliness of a particular tool or method, the user will have a greater intention to use the tool if they consider that tool to be relatively user-friendly (Barat et al., 2009). The findings also show the importance of perceived convenience in enhancing the intention of users to use online learning systems through the moderating role of PU.

According to the findings, the PU predicted the intention to use online education systems, and the effect size was strong. If people expect the information and educational services provided by online systems to improve their learning, then they will likely be more inclined to use these systems. These results are consistent with many past studies (Calisir et al., 2014; Alqahtani et al., 2016; Tarhini et al., 2017; Fallah Haghighi et al., 2018; Al-Rahmi et al., 2019; Hariguna, 2019; Salloum et al., 2019; Tao et al., 2019). Our study is also consistent with Li et al. (2012) and Tao et al. (2019) who confirmed the indirect effect of PU on intention through attitude. The study by Park et al. (2012) on the acceptance of collaborative learning technologies also confirms this finding.

In line with past research we found that PEOU could strongly affect PU (Luarn and Lin, 2005; Lee, 2009; Park et al., 2012; Kim, 2014; Bhatiasevi and Yoopetch, 2015; Alqahtani et al., 2016; Chang et al., 2017; Ebrahimi et al., 2018; Joo et al., 2018; Tao et al., 2019). Students will find online learning useful and effective if they are comfortable using the Internet and the learning system and as long as they do not encounter any difficulties or problems with them.

Based on our findings, output quality and Internet self-efficacy had a significant indirect effect on intention in addition to the main TAM constructs. While Internet anxiety did not have a significant indirect effect on intention, it did have has a direct positive effect on PU. A review of studies shows there to be contradictions about the impact of Internet anxiety on PU. Zhang (2005) confirmed the positive effect of Internet anxiety on PU with respect to telecommunications employees using the Internet. Other research findings reported the negative effect of Internet anxiety on PU (Park et al., 2012; Purnomo and Lee, 2013).

Internet anxiety also had a negative effect on PEOU with a moderate effect size. The more students experience unrest, panic, fear, and anxiety in general when using the Internet, the more difficult it is to use a learning system. The negative effect of anxiety on PEOU has also been confirmed in research (Karaali et al., 2011; Chen and Tseng, 2012; Park et al., 2012; Ali et al., 2013; Calisir et al., 2014; Al-Gahtani, 2016).

The findings showed that output quality is a direct determinant of PU and PEOU. Perceptions of output quality results from cognitive instrumental processes (Venkatesh and Davis, 2000). In this study, determining output quality means comparing the result of learning online with the perceived quality of learning in traditional classrooms. In fact, students' assessment of the quality of outcome of online classes may be different from the perceived output quality of learning in traditional classrooms. Tao et al. (2019) found that perceived quality affects the PU of massive open online courses. Consistent with our results, the positive effect of output quality on PU (Ahn et al., 2004; Vijayasarathy, 2004; Ahn et al., 2007) and on PEOU (Lee, 2010) was confirmed. Al-Nuaimi et al. (2022) also found that learning management technical system quality has demonstrated the most critical impact on the PU.

According to our findings, Internet self-efficacy indirectly affected attitudes and intention. The research of Isaac et al. (2017) found Internet self-efficacy to be a strong factor in the intention of employees in Yemeni government institutions to use online systems. However, the direct effect of self-efficacy on PU was only approaching significance (p = 0.055) while effect of self-efficacy on PEOU was moderate. Here, Internet self-efficacy refers to the belief of a person in their ability to use the Internet. In this regard, Tsai et al. (2014) showed that system self-efficacy can affect the PU and PEOU of telehealth systems. Similarly, Arpaci et al. (2019) found that self efficacy has direct effect on the PEOU.

Faqih (2013) examined consumer intention in Jordan and confirmed that the influence of Internet self-efficacy on PEOU was positive and that strong Internet self-efficacy affects PU positively with 99% confidence. Research also confirms the positive and strong effect of Internet self-efficacy on PU and PEOU (Park, 2009; Isaac et al., 2017).

6. Theoretical and practical implications

Theoretically, the current study provides new evidence to the existing literature supporting the TAM by confirming the significant positive effects of PU and PEOU on attitudes toward online learning systems. This study suggests that the intention to use online learning systems can be increased by developing the constructs of internet anxiety, internet effectiveness, and output quality.

The outcomes of the current study provide valuable knowledge and offer a deeper understanding of external factors and provide useful practical suggestions for policymakers, professionals, developers, and designers in the effective use of online learning systems during and beyond Covid-19. The findings may have implications for organizations planning to design and implement online learning systems.

As PU is a direct predictor of behavioral intention, managers can use this to enhance people's beliefs about how online learning systems can improve their learning performance and thereby increase the intention of students to use them even in non-emergency health situations. Specifically, students need to be made aware of how online learning systems will benefit them and improve their performance. In this research, PU has been shown with criteria such as the value of the online learning experience, the objectivity of the online learning results, and the minimizing of educational inequality. Managers should review communication materials (e.g., websites and brochures) designed to publicize these systems and make sure the utility in individual level such as increased learning and social benefits of online learning such as reducing educational is emphasized.

As attitude was the most important predictor of intention, and PU and PEOU both directly and indirectly affected intention, educators need to ensure that online learning is beneficial to improving student learning outcomes. Superior teaching techniques should provide for greater success. The findings imply administrators and online educators must devote resources to ensuring online education is designed appropriately and effectively. It is not enough to simply place an established class online; online courses need to be designed using best practices for this type of educational modality.

Considering the effect of output quality on PU and PEOU, there should be a statistical comparison of the learning performance of online and traditional learning systems. This analysis can also ensure quality of online learning offerings. Additionally, these findings can be also shared with students to help them make decisions regarding the usefulness of online learning. Providing information in this area can improve people's understanding of learning quality. The use of user-friendly methods *via* a variety of educational techniques, including videos, photos, text news, etc. seems to improve people's understanding of learning quality. Moreover, given the negative impact of Internet anxiety on PEOU, students need to be specifically educated on Internet use. Educational institutions offering online learning should consider offering seminar type courses to improve comfort with Internet use, or embedding this type of training into the beginning of online courses.

This study provides a strong insight for the government and the Ministry of Education to provide all the educational material related to the internet and computers to the students. For this purpose, it is suggested that students be provided with the purchase of powerful modems and wireless and high-speed internet packages on Sim-cards.

The findings strongly suggest, however, that online learning systems should feature ease of use and simplify the learning process. Therefore, it is important to improve features of online learning system, helping users complete tasks with the less time and effort. Efforts should be made to guarantee that online learning systems are of a simple yet advanced design that allow the inclusion of a variety of educational methods. Therefore, it is recommended to use easy and user-friendly platforms with the ability to share all kinds of files and videos, with the ability to provide interactive presentations in the simplest possible mode, for training.

Perceived quality output also has an influence on the intent to usage online education. The disadvantages of unexpected operation of online education could lead government, regulators and other practitioners to advance the online education as well as increase the stdent leaning. Clearly, respected criticism from learner and educator should be considered for future improvment since the learning methods have extremely reformed due to the Covid-19 pandamic. The criticisms could inspire the educational institute to improve the quality of platform.

7. Conclusion and limitations

This study has several important conclusions: First, PU and PEOU must be taken seriously in order to bring a successful online learning system into effect. Students' attitudes to using online learning systems were the strongest predictors of behavioral intention. The findings confirm the direct and indirect impact of output quality on all endogenous constructs including PU, PEOU, attitude, and intention. As university administrators need to improve the learning performance of individuals, output quality needs to be made an urgent priority in learning system projects. If users perceive that virtual learning systems will enhance learning performance to a greater extent than traditional learning systems do, they will have higher PU and therefore look more favorably on online learning systems and be more inclined to use them. Output quality was the strongest predictor of PU and PEOU. Educational planners and designers thus need to pay adequate attention to this important element.

The present study, like other studies, has limitations that need to be considered when this topic is tackled in the future. First, this study was conducted with agricultural students at a limited number of universities in Iran, where practical and experimental courses were lacking, and in which various online education systems were used. Our findings were extracted from the analysis of this data. Generalizations to other students and other universities should thus be made with caution.

Second, this study integrated the three key constructs of output quality, Internet anxiety, and Internet self-efficacy into TAM. Further studies have shown that other constructs can be used to improve the prediction of the intention to use online learning. Which suggests additional variables could be added in future studies.

One of the important limitations of this study was how the questions were assessed. In this study, items were measured as self-reported. To address the potential influence of common method bias we used a range of different items to measure our sample perception. Furtherrmore we used positive and negative (Reverse items) items and finally adapting our measures from reliable and valid instruments helps us in this regard. However, it is suggested, that other methods of data collection, such as interviews and focus groups, be used in other studies beside structured questionnaires.

This research was also cross-sectional and evaluated people's perceptions and intention at a specific time, and particularly during the peak of a global pandemic. People's perceptions will, however, change over time, as their experience increases (Venkatesh et al., 2003). Future researchers are advised to consider using longitudinal studies.

Data availability statement

The original contributions presented in the study are included in the article/Supplementary material, further inquiries can be directed to the corresponding author/s.

Ethics statement

Ethical review and approval was not required for the study on human participants in accordance with the local legislation and institutional requirements. All procedures performed in studies involving human participants were in accordance with the ethical standards of the institutional and/or national research committee and with the 1964 Helsinki declaration and its later amendments or comparable ethical standards. Written informed consent was obtained from all subjects involved in the study.

Author contributions

TZ: conceptualization, methodology, software, formal analysis, investigation, writing – original draft, and writing – review and editing. SH: methodology, software, validation, formal analysis,

References

Abbasi, M. S., Chandio, F. H., Soomro, A. F., and Shah, F. (2011). Social influence, voluntariness, experience and the internet acceptance: an extension of technology acceptance model within a south-Asian country context. *J. Enterp. Inf. Manag.* 24, 30–52. doi: 10.1108/17410391111097410

Aboagye, E., Yawson, J. A., and Appiah, K. N. (2021). COVID-19 and E-learning: the challenges of students in tertiary institutions. *Soc. Educ. Res.* 2, 1–8. doi: 10.37256/ser.212021422

Agarwal, R., and Prasad, J. (1999). Are individual differences germane to the acceptance of new information technologies? *Decis. Sci.* 30, 361–391. doi: 10.1111/j.1540-5915.1999. tb01614.x

Ahn, T., Ryu, S., and Han, I. (2004). The impact of the online and offline features on the user acceptance of internet shopping malls. *Electron. Commer. Res. Appl.* 3, 405–420. doi: 10.1016/j.elerap.2004.05.001

Ahn, T., Ryu, S., and Han, I. (2007). The impact of web quality and playfulness on user acceptance of online retailing. *Inf. Manag.* 44, 263–275. doi: 10.1016/j.im.2006.12.008

Ajzen, I. (1985). "From intentions to actions: a theory of planned behavior" in *Action Control*. eds. I. J. Kuhl and J. Beckmann (Berlin, Heidelberg: Springer Berlin Heidelberg), 11–39.

Ajzen, I., and Fishbein, M. (1975). A Bayesian analysis of attribution processes. *Psychol. Bull.* 82, 261–277. doi: 10.1037/h0076477

Ajzen, I., and Fishbein, M., (1980). Understanding Attitudes and Predicting Social Behavior. Englewood Cliffs, NJ: Prentice Hall.

Al-Adwan, A. S. (2020). Investigating the drivers and barriers to MOOCs adoption: the perspective of TAM. *Educ. Inf. Technol.* 25, 5771–5795. doi: 10.1007/s10639-020-10250-z

Al-Adwan, A. S., Albelbisi, N. A., Hujran, O., Al-Rahmi, W. M., and Alkhalifah, A. (2021). Developing a holistic success model for sustainable e-learning: a structural equation modeling approach. *Sustainability* 13:9453. doi: 10.3390/su13169453

Al-Adwan, A. S., Nofal, M., Akram, H., Albelbisi, N. A., and Al-Okaily, M. (2022). Towards a sustainable adoption of E-learning systems: the role of self-directed learning. *J. Inf. Technol. Educ. Res.* 21, 245–267. doi: 10.28945/4980

Al-Alak, B. A., and Alnawas, I. A. (2011). Measuring the acceptance and adoption of e-learning by academic staff. *Knowl. Manage. e-Learn. Int. J.* 3, 201–221.

Al-Emran, M., and Granić, A. (2021). "Is it still valid or outdated? A bibliometric analysis of the technology acceptance model and its applications from 2010 to 2020" in *Recent Advances in Technology Acceptance Models and Theories* (Cham: Springer), 1–12.

Al-Emran, M., Granić, A., Al-Sharafi, M. A., Ameen, N., and Sarrab, M. (2020). Examining the roles of students' beliefs and security concerns for using smartwatches in higher education. *J. Enterp. Inf. Manag.* 34, 1229–1251. doi: 10.1108/JEIM-02-2020-0052

Alenezi, A. R., and Karim, A. (2010). An empirical investigation into the role of enjoyment, computer anxiety, computer self-efficacy and internet experience in influencing

writing – original draft, and writing – review and editing. MY: conceptualization, methodology, software, validation, formal analysis, investigation, writing – original draft, writing – review and editing, visualization, and supervision. NK and LW: formal analysis, writing – review and editing. All authors contributed to the article and approved the submitted version.

Conflict of interest

The authors declare that the research was conducted in the absence of any commercial or financial relationships that could be construed as a potential conflict of interest.

Publisher's note

All claims expressed in this article are solely those of the authors and do not necessarily represent those of their affiliated organizations, or those of the publisher, the editors and the reviewers. Any product that may be evaluated in this article, or claim that may be made by its manufacturer, is not guaranteed or endorsed by the publisher.

the students' intention to use e-learning: a case study from Saudi Arabian governmental universities. *Turk. Online J. Educ. Technol.* 9, 22–34.

Al-Fraihat, D., Joy, M., and Sinclair, J. (2020). Evaluating E-learning systems success: an empirical study. *Comput. Hum. Behav.* 102, 67–86. doi: 10.1016/j.chb.2019.08.004

Al-Gahtani, S. S. (2016). Empirical investigation of e-learning acceptance and assimilation: a structural equation model. *Appl. Comput. Inf.* 12, 27–50. doi: 10.1016/j. aci.2014.09.001

Ali, H., Ahmed, A. A., Tariq, T. G., and Safdar, H. (2013). Second Life (SL) in Education: The Intensions to Use at University of Bahrain. In 2013 Fourth International Conference on e-Learning "Best Practices in Management, Design and Development of e-Courses: Standards of Excellence and Creativity" (pp. 205–215). IEEE.

Alksasbeh, M., Abuhelaleh, M., and Almaiah, M. (2019). Towards A Model of Quality Features for Mobile Social Networks Apps in Learning Environments: An Extended Information System Success Model. 13, 75–93. doi: 10.3991/ijim.v13i05.9791

Almaiah, M. A., and Alamri, M. M. (2018). Proposing a new technical quality requirement for mobile learning applications. *J. Theor. Appl. Inf. Technol.* 96, 6955–6968.

Almaiah, M. A., and Al-Khasawneh, A. (2020). Investigating the main determinants of mobile cloud computing adoption in university campus. *Educ. Inf. Technol.* 25, 3087–3107. doi: 10.1007/s10639-020-10120-8

Almaiah, M. A., and Alismaiel, O. A. (2019). Examination of factors influencing the use of mobile learning system: An empirical study. *Education and Information Technologies* 24, 885–909. doi: 10.1007/s10639-018-9810-7

Al-Nuaimi, M. N., Al Sawafi, O. S., Malik, S. I., Al-Emran, M., and Selim, Y. F. (2022). Evaluating the actual use of learning management systems during the covid-19 pandemic: an integrated theoretical model. *Interact. Learn. Environ.*, 1–26. doi: 10.1080/10494820.2022.2055577

Alqahtani, A., Crowder, R., and Wills, G. (2016). Barriers to the adoption of EHR systems in the kingdom of Saudi Arabia: an exploratory study using a systematic literature review. *J. Health Inf. Dev. Ctries* 11, 1–23.

Al-Rahmi, W. M., Yahaya, N., Aldraiweesh, A. A., Alamri, M. M., Aljarboa, N. A., Alturki, U., et al. (2019). Integrating technology acceptance model with innovation diffusion theory: an empirical investigation on students' intention to use E-learning systems. *IEEE Access* 7, 26797–26809. doi: 10.1109/ACCESS.2019.2899368

Amghani, M. S., Savari, M., and Choobchian, S. (2022). Vulnerability assessment of Iran's rural-farmer households during COVID-19 pandemic. *Front. Public Health* 10:994922. doi: 10.3389/fpubh.2022.994922

Anderson, R. M., Heesterbeek, H., Klinkenberg, D., and Hollingsworth, T. D. (2020). How will country-based mitigation measures influence the course of the COVID-19 epidemic? *Lancet* 395, 931–934. doi: 10.1016/S0140-6736(20)30567-5 Arpaci, I. (2017). The role of self-efficacy in predicting use of distance education tools and learning management systems. *Turk. Online J. Dist. Educ.* 18, 52–62. doi: 10.17718/tojde.285715

Arpaci, I., Durdu, P. O., and Mutlu, A. (2019). The role of self-efficacy and perceived enjoyment in predicting computer engineering Students' continuous use intention of scratch. *Int. J. e-Adopt.* 11, 1–12. doi: 10.4018/IJEA.2019070101

Arpaci, I., Karatas, K., Kusci, I., and Al-Emran, M. (2022). Understanding the social sustainability of the Metaverse by integrating UTAUT2 and big five personality traits: a hybrid SEM-ANN approach. *Technol. Soc.* 71:102120. doi: 10.1016/j.techsoc.2022.102120

Ataei, P., Ghadermarzi, H., Karimi, H., and Norouzi, A. (2021a). The process of adopting entrepreneurial behaviour: evidence from agriculture students in Iran. *Innov. Educ. Teach. Int.* 58, 340–350. doi: 10.1080/14703297.2020.1734476

Ataei, P., Gholamrezai, S., Movahedi, R., and Aliabadi, V. (2021b). An analysis of farmers' intention to use green pesticides: the application of the extended theory of planned behavior and health belief model. *J. Rural. Stud.* 81, 374–384. doi: 10.1016/j. jrurstud.2020.11.003

Bandura, A. (1986). *Prentice-Hall Series in Social Learning Theory*. Social Foundations of Thought and Action: A Social Cognitive Theory. Englewood Cliffs, NJ.

Barat, S., Rajamma, R. K., Zolfagharian, M. A., and Ganesh, G. (2009). Student course perceptions: a perceived-ease-of-use--perceived-usefulness framework. *J. Adv. Mark. Educ.* 15:149.

Bhatiasevi, V., and Yoopetch, C. (2015). The determinants of intention to use electronic booking among young users in Thailand. *J. Hosp. Tour. Manag.* 23, 1–11. doi: 10.1016/j. jhtm.2014.12.004

Briz-Ponce, L., Pereira, A., Carvalho, L., Juanes-Méndez, J. A., and García-Peñalvo, F. J. (2017). Learning with mobile technologies-students' behavior. *Comput. Hum. Behav.* 72, 612–620. doi: 10.1016/j.chb.2016.05.027

Buabeng-Andoh, C. (2021). Exploring university students' intention to use mobile learning: a research model approach. *Educ. Inf. Technol.* 26, 241–256. doi: 10.1007/s10639-020-10267-4

Calisir, F., Altin Gumussoy, C., Bayraktaroglu, A. E., and Karaali, D. (2014). Predicting the intention to use a web-based learning system: perceived content quality, anxiety, perceived system quality, image, and the technology acceptance model. *Hum. Fact. Ergon. Manufact. Ser. Ind.* 24, 515–531. doi: 10.1002/hfm.20548

Chang, C. T., Hajiyev, J., and Su, C. R. (2017). Examining the students' behavioral intention to use e-learning in Azerbaijan? The general extended technology acceptance model for e-learning approach. *Comput. Educ.* 111, 128–143. doi: 10.1016/j. compedu.2017.04.010

Chang, S. J., and Im, E. O. (2014). A path analysis of internet health information seeking behaviors among older adults. *Geriatr. Nurs.* 35, 137–141. doi: 10.1016/j. gerinurse.2013.11.005

Chen, H. R., and Tseng, H. F. (2012). Factors that influence acceptance of web-based e-learning systems for the in-service education of junior high school teachers in Taiwan. *Eval. Program Plann.* 35, 398–406. doi: 10.1016/j.evalprogplan.2011.11.007

Cheng, Y. M. (2011). Antecedents and consequences of e-learning acceptance. *Inf. Syst. J.* 21, 269–299. doi: 10.1111/j.1365-2575.2010.00356.x

Chin, W. W. (1998). The partial least squares approach to structural equation modeling. *Mod. Methods Bus. Res.* 295, 295–336.

Chung, J. E., Park, N., Wang, H., Fulk, J., and McLaughlin, M. (2010). Age differences in perceptions of online community participation among non-users: an extension of the technology acceptance model. *Comput. Hum. Behav.* 26, 1674–1684. doi: 10.1016/j. chb.2010.06.016

Chung, E., Subramaniam, G., and Dass, L. C. (2020). Online learning readiness among university students in Malaysia amidst COVID-19. *Asian J. Univ. Educ.* 16, 45–58. doi: 10.24191/ajue.v16i2.10294

Compeau, D. R., and Higgins, C. A. (1995). Computer self-efficacy: development of a measure and initial test. *MIS Q.* 19, 189–211. doi: 10.2307/249688

Davis, F. D. (1989). PU, PEOU, and user acceptance of information technology. *MIS Q.* 13, 319–340. doi: 10.2307/249008

Davis, F. D. (1993). User acceptance of information technology: system characteristics, user perceptions and behavioral impacts. *Int. J. Man Machine Stud.* 38, 475–487. doi: 10.1006/imms.1993.1022

Davis, F. D., Bagozzi, R. P., and Warshaw, P. R. (1992). Extrinsic and intrinsic motivation to use computers in the workplace 1. *J. Appl. Soc. Psychol.* 22, 1111–1132. doi: 10.1111/j.1559-1816.1992.tb00945.x

Demoulin, N. T., and Coussement, K. (2020). Acceptance of text-mining systems: the signaling role of information quality. *Inf. Manag.* 57:103120. doi: 10.1016/j.im.2018.10.006

Eastin, M. S., and LaRose, R. (2000). Internet self-efficacy and the psychology of the digital divide. *J. Comput.-Mediat. Commun.* 6:JCMC611. doi: 10.1111/j.1083-6101.2000. tb00110.x

Ebrahimi, S., Moeinikia, M., and Zahed Babelan, A. (2018). Simple and multiple relationships among PEOU and PU with E-learning acceptance in universities' instructors. *Q. Iran. Distance Educ. J.* 1, 45–50.

Eom, S. B., Wen, H. J., and Ashill, N. (2006). The determinants of students' perceived learning outcomes and satisfaction in university online education: an empirical investigation. *Decis. Sci. J. Innov. Educ.* 4, 215–235. doi: 10.1111/j.1540-4609.2006.00114.x

Fallah Haghighi, N., Mahmoudi, M., and Bijani, M. (2018). Barriers to entrepreneurship development in Iran's higher education: a qualitative case study. *Interchange* 49, 353–375. doi: 10.1007/s10780-018-9330-9

Fallon, C., and Brown, S. (2002). *E-Learning Standards: A Guide to Purchasing, Developing, and Deploying Standards-Conformant e-Learning, Routledge, Taylor & Francis group.*

Faqih, K. M. (2013). Exploring the influence of perceived risk and internet self-efficacy on consumer online shopping intentions: perspective of technology acceptance model. *Int. Manage. Rev.* 9, 67–77.

Fornell, C., and Bookstein, F. L. (1982). Two structural equation models: LISREL and PLS applied to consumer exit-voice theory. J. Mark. Res. 19, 440–452. doi: 10.1177/002224378201900406

Fornell, C., and Larcker, D. F. (1981). Evaluating structural equation models with unobservable variables and measurement error. *J. Mark. Res.* 18, 39–50. doi: 10.1177/002224378101800104

Gewin, V. (2020). Five tips for moving teaching online as COVID-19 takes hold. *Nature* 580, 295–296. doi: 10.1038/d41586-020-00896-7

Ghafourifard, M. (2020). The promotion of virtual education in Iran: the potential which turned into reality by coronavirus. *Iran. J. Med. Educ.* 20, 33–34.

Hair, J. F., Risher, J. J., Sarstedt, M., and Ringle, C. M. (2019). When to use and how to report the results of PLS-SEM. *Eur. Bus. Rev.* 31, 2–24. doi: 10.1108/EBR-11-2018-0203

Hallaj, Z., Bijani, M., Abbasi, E., Valizadeh, N., and Mohammadi, M. (2022). Tourism development during the pandemic of coronavirus (COVID-19): evidence from Iran. *Front. Public Health* 10:881381. doi: 10.3389/fpubh.2022

Han, J. H., and Sa, H. J. (2022). Acceptance of and satisfaction with online educational classes through the technology acceptance model (TAM): the COVID-19 situation in Korea. *Asia Pac. Educ. Rev.* 23, 403–415. doi: 10.1007/s12564-021-09716-7

Hanif, A., Jamal, F. Q., and Imran, M. (2018). Extending the technology acceptance model for use of e-learning systems by digital learners. *IEEE Access* 6, 73395–73404. doi: 10.1109/ACCESS.2018.2881384

Hariguna, T. (2019). Assessing students' continuance intention in using multimedia online learning. *Telkomnika* 17, 187–193. doi: 10.12928/telkomnika.v17i1.10328

Henseler, J., Ringle, C. M., and Sinkovics, R. R. (2009). *The use of partial least squares path modeling in international marketing*. In New Challenges to International Marketing. Emerald: Emerald Group Publishing Limited.

Hernández-García, A., and Iglesias-Pradas, S. (2012). Desarrollo de un Modelo Unificado de Adopción del Comercio Electrónico Entre Empresas y Consumidores Finales. Aplicación al Mercado Español (Doctoral Dissertation, Telecomunicacion).

Huang, Y. M. (2016). The factors that predispose students to continuously use cloud services: social and technological perspectives. *Comput. Educ.* 97, 86–96. doi: 10.1016/j. compedu.2016.02.016

Hussein, Z. (2017). Leading to intention: the role of attitude in relation to technology acceptance model in e-learning. *Proce. Comput. Sci.* 105, 159–164. doi: 10.1016/j. procs.2017.01.196

Ibili, E., Resnyansky, D., and Billinghurst, M. (2019). Applying the technology acceptance model to understand maths teachers' perceptions towards an augmented reality tutoring system. *Educ. Inf. Technol.* 24, 2653–2675. doi: 10.1007/s10639-019-09925-z

Idris, F. A. A., and Osman, Y. B. (2015). Challanges facing the implementation of e-learning at University of Gezira According to View of Staff Members. In 2015 Fifth International Conference on e-Learning (ECONF) (pp. 336–348). IEEE.

Igbaria, M., and Parasuraman, S. (1989). A path analytic study of individual characteristics, computer anxiety and attitudes toward microcomputers. *J. Manag.* 15, 373–388. doi: 10.1177/014920638901500302

Isaac, O., Abdullah, Z., Ramayah, T., and Mutahar, A. M. (2017). Internet usage within government institutions in Yemen: an extended technology acceptance model (TAM) with internet self-efficacy and performance impact. *Sci. Int.* 29, 737–747.

Ji, Z., Yang, Z., Liu, J., and Yu, C. (2019). Investigating users' continued usage intentions of online learning applications. *Information* 10:198. doi: 10.3390/info10060198

Joo, Y. J., So, H. J., and Kim, N. H. (2018). Examination of relationships among students' self-determination, technology acceptance, satisfaction, and continuance intention to use K-MOOCs. *Comput. Educ.* 122, 260–272. doi: 10.1016/j.compedu.2018.01.003

Joosten, T., and Cusatis, R. (2020). Online learning readiness. Am. J. Dist. Educ. 34, 180–193. doi: 10.1080/08923647.2020.1726167

Karaali, D., Gumussoy, C. A., and Calisir, F. (2011). Factors affecting the intention to use a web-based learning system among blue-collar workers in the automotive industry. *Comput. Hum. Behav.* 27, 343–354. doi: 10.1016/j.chb.2010.08.012

Karataş, K., Arpaci, I., and Yildirim, Y. (2022). Predicting the culturally responsive teacher roles with cultural intelligence and self-efficacy using machine learning classification algorithms. *Educ. Urban Soc.* 00131245221087999:0013124522108799. doi: 10.1177/00131245221087999

Khayun, V., and Ractham, P. (2011). Measuring e-Excise Tax Success Factors: Applying the DeLone and McLean Information Systems Success Model. In 2011 44th Hawaii International Conference on System Sciences (pp. 1–10). IEEE.

Kim, S. H. (2014). A study on adoption factors of Korean smartphone users: a focus on TAM (technology acceptance model) and UTAUT (unified theory of acceptance and use of technology). *Adv. Sci. Tech.* 57, 27–30. doi: 10.14257/astl.2014.57.07

Kim, K. J., Liu, S., and Bonk, C. J. (2005). Online MBA students' perceptions of online learning: benefits, challenges, and suggestions. *Internet High. Educ.* 8, 335–344. doi: 10.1016/j.iheduc.2005.09.005

Kuo, Y. C. (2010). Interaction, Internet Self-Effifficacy, and Self-Regulated Learning as Predictors of Student Satisfaction in Distance Education Courses. All Graduate Theses and Dissertations, 741. doi: 10.1016/j.iheduc.2013.10.001

Kuo, Y. C., Walker, A. E., Schroder, K. E., and Belland, B. R. (2014). Interaction, internet self-efficacy, and self-regulated learning as predictors of student satisfaction in online education courses. *Internet High. Educ.* 20, 35–50. doi: 10.1016/j. iheduc.2013.10.001

Lahordi, N. (2021). Internet speed in Iran and the world. *Donya e-Eqtesad.* 5099: 3736527

Lee, M. C. (2009). Factors influencing the adoption of internet banking: an integration of TAM and TPB with perceived risk and perceived benefit. *Electron. Commer. Res. Appl.* 8, 130–141. doi: 10.1016/j.elerap.2008.11.006

Lee, J. W. (2010). Online support service quality, online learning acceptance, and student satisfaction. *Internet High. Educ.* 13, 277–283. doi: 10.1016/j.iheduc.2010.08.002

Lee, Y. H., Hsiao, C., and Purnomo, S. H. (2014). An empirical examination of individual and system characteristics on enhancing e-learning acceptance. *Australas. J. Educ. Technol.* 30:381. doi: 10.14742/ajet.381

Lee, Y. H., Hsieh, Y. C., and Chen, Y. H. (2013). An investigation of employees' use of e-learning systems: applying the technology acceptance model. *Behav. Inform. Technol.* 32, 173–189. doi: 10.1080/0144929X.2011.577190

Lee, Y. H., Hsieh, Y. C., and Ma, C. Y. (2011). A model of organizational employees' e-learning systems acceptance. *Knowl.-Based Syst.* 24, 355–366. doi: 10.1016/j. knosys.2010.09.005

Lee, Y., Kozar, K. A., and Larsen, K. R. (2003). The technology acceptance model: past, present, and future. *Commun. Assoc. Inf. Syst.* 12:50. doi: 10.17705/1CAIS.01250

Lee, D., Lee, W. H., Seo, H. G., Oh, B. M., Lee, J. C., and Kim, H. C. (2020). Online learning for the hyoid bone tracking during swallowing with neck movement adjustment using semantic segmentation. *IEEE Access* 8, 157451–157461. doi: 10.1109/ACCESS.2020.3019532

Lee, W. H., Lin, C. W., and Shih, K. H. (2018). A technology acceptance model for the perception of restaurant service robots for trust, interactivity, and output quality. *Int. J. Mob. Commun.* 16, 361–376. doi: 10.1504/IJMC.2018.092666

Lee, B. C., Yoon, J. O., and Lee, I. (2009). Learners' acceptance of e-learning in South Korea: theories and results. *Comput. Educ.* 53, 1320–1329. doi: 10.1016/j. compedu.2009.06.014

Li, Y., Duan, Y., Fu, Z., and Alford, P. (2012). An empirical study on behavioural intention to reuse e-learning systems in rural China. *Br. J. Educ. Technol.* 43, 933–948. doi: 10.1111/j. 1467-8535.2011.01261.x

Luarn, P., and Lin, H. H. (2005). Toward an understanding of the behavioral intention to use mobile banking. *Comput. Hum. Behav.* 21, 873–891. doi: 10.1016/j. chb.2004.03.003

Mahdavian, M., Wingreen, S. C., and Zealand, N. E. W. (2016). The influence of key users †tm skills on ERP success. J. Inf. Technol. Manag 27, 48–64.

Mailizar, M., Burg, D., and Maulina, S. (2021). Examining university students' behavioural intention to use e-learning during the COVID-19 pandemic: an extended TAM model. *Educ. Inf. Technol.* 26, 7057–7077. doi: 10.1007/s10639-021-10557-5

McMaster, T., and Wastell, D. (2005). Diffusion–or delusion? Challenging an IS research tradition. *Inf. Technol. People* 18, 383–404. doi: 10.1108/09593840510633851

Moradhaseli, S., Ataei, P., Karimi, H., and Hajialiany, S. (2022). Typology of Iranian farmers' vulnerability to the COVID-19 outbreak. *Front. Public Health* 10:1018406. doi: 10.3389/fpubh.2022.1018406

Mouakket, S., and Bettayeb, A. M. (2016). Factors Affecting University Instructors' Continuance Intention to Use Learning Management Systems: The Blackboard System Case. In International Conference on Enterprise Information Systems (Vol. 3, pp. 215–222). SCITEPRESS.

Naveed, Q. N., Muhammed, A., Sanober, S., Qureshi, M. R. N., and Shah, A. (2017). Barriers effecting successful implementation of E-learning in Saudi Arabian universities. *Int. J. Emerg. Technol. Learn.* 12:94. doi: 10.3991/ijet.v12i06.7003

Nunnally, J. C. (1978). Psychometric Theory. 2nd Edn, McGraw-Hill Companies.

Padilla-Meléndez, A., del Aguila-Obra, A. R., and Garrido-Moreno, A. (2013). Perceived playfulness, gender differences and technology acceptance model in a blended learning scenario. *Comput. Educ.* 63, 306–317. doi: 10.1016/j.compedu.2012.12.014

Pal, D., and Vanijja, V. (2020). Perceived usability evaluation of Microsoft teams as an online learning platform during COVID-19 using system usability scale and technology acceptance model in India. *Child Youth Serv. Rev.* 119:105535. doi: 10.1016/j. childyouth.2020.105535

Park, S. Y. (2009). An analysis of the technology acceptance model in understanding university students' behavioral intention to use e-learning. J. Educ. Technol. Soc. 12, 150–162.

Park, S. Y., Nam, M. W., and Cha, S. B. (2012). University students' behavioral intention to use mobile learning: evaluating the technology acceptance model. *Br. J. Educ. Technol.* 43, 592–605. doi: 10.1111/j.1467-8535.2011.01229.x

Phillips, L. A., Calantone, R., and Lee, M. T. (1994). International technology adoption: behavior structure, demand certainty and culture. *J. Bus. Ind. Mark.* 9, 16–28. doi: 10.1108/08858629410059762

Purnomo, S. H., and Lee, Y. H. (2013). E-learning adoption in the banking workplace in Indonesia: an empirical study. *Inf. Dev.* 29, 138–153. doi: 10.1177/0266666912448258

Rahman, M. A., Qi, X., and Islam, M. (2016). Banking access for the poor: adoption and strategies in rural areas of Bangladesh. *J. Econ. Financ. Stud.* 4, 01–10. doi: 10.18533/jefs. v4i3.236

Rahmawati, R. N. (2019). Self-efficacy and use of e-learning: a theoretical review technology acceptance model (TAM). Am. J. Hum. Soc. Sci. Res. 3, 41–55.

Raza, S. A., Qazi, W., Khan, K. A., and Salam, J. (2021). Social isolation and acceptance of the learning management system (LMS) in the time of COVID-19 pandemic: an expansion of the UTAUT model. *J. Educ. Comput. Res.* 59, 183–208. doi: 10.1177/0735633120960421

Revythi, A., and Tselios, N. (2019). Extension of technology acceptance model by using system usability scale to assess behavioral intention to use e-learning. *Educ. Inf. Technol.* 24, 2341–2355. doi: 10.1007/s10639-019-09869-4

Romanov, K., and Nevgi, A. (2006). Learning outcomes in medical informatics: comparison of a WebCT course with ordinary web site learning material. *Int. J. Med. Inform.* 75, 156–162. doi: 10.1016/j.ijmedinf.2005.06.004

Sahu, P. (2020). Closure of universities due to coronavirus disease 2019 (COVID-19): impact on education and mental health of students and academic staff. *Cureus* 12:e7541. doi: 10.7759/cureus.7541

Salloum, S. A., Alhamad, A. Q. M., Al-Emran, M., Monem, A. A., and Shaalan, K. (2019). Exploring students' acceptance of e-learning through the development of a comprehensive technology acceptance model. *IEEE Access* 7, 128445–128462. doi: 10.1109/ ACCESS.2019.2939467

Sánchez-Prieto, J. C., Olmos-Migueláñez, S., and García-Peñalvo, F. J. (2017). MLearning and pre-service teachers: an assessment of the behavioral intention using an expanded TAM model. *Comput. Hum. Behav.* 72, 644–654. doi: 10.1016/j.chb.2016.09.061

Sarcheshmeh, E. E., Bijani, M., and Sadighi, H. (2018). Adoption behavior towards the use of nuclear technology in agriculture: a causal analysis. *Technol. Soc.* 55, 175–182. doi: 10.1016/j.techsoc.2018.08.001

Savari, M., Naghibeiranvand, F., and Asadi, Z. (2022). Modeling environmentally responsible behaviors among rural women in the forested regions in Iran. *Global Ecol. Conserv.* 35:e02102. doi: 10.1016/j.gecco.2022.e02102

Savari, M., Zhoolideh, M., and Khosravipour, B. (2021). Explaining pro-environmental behavior of farmers: a case of rural Iran. *Curr. Psychol.*, 1–19. doi: 10.1007/s12144-021-02093-9

Shahangian, S. A., Tabesh, M., and Yazdanpanah, M. (2021). How can socio-psychological factors be related to water-efficiency intention and behaviors among Iranian residential water consumers? *J. Environ. Manag.* 288:112466. doi: 10.1016/j.jenvman.2021.112466

Sharifi, M., Fathabadi, J., Shokri, O., and Pakdaman, S. (2019). The experience of e-learning in the educational system of Iran: meta-analysis of the effectiveness of e-learning in comparison to face-to-face education. *J. Res. Sch. Virtual Learn.* 7, 9–24. (in persian)

Shawai, Y. G., and Almaiah, M. A. (2018). Malay language mobile learning system (MLMLS) using NFC technology. *Int. J. Educ. Manage. Eng.* 8, 1–7. doi: 10.5815/ ijeme.2018.02.01

Sinclair, P. M., Kable, A., Levett-Jones, T., and Booth, D. (2016). The effectiveness of internet-based e-learning on clinician behaviour and patient outcomes: a systematic review. *Int. J. Nurs. Stud.* 57, 70–81. doi: 10.1016/j.ijnurstu.2016.01.011

Sivo, S. A., Ku, C. H., and Acharya, P. (2018). Understanding how university student perceptions of resources affect technology acceptance in online learning courses. *Australas. J. Educ. Technol.* 34:2806. doi: 10.14742/ajet.2806

Sukendro, S., Habibi, A., Khaeruddin, K., Indrayana, B., Syahruddin, S., Makadada, F. A., and Hakim, H. (2020). Using an extended Technology Acceptance Model to understand students' use of e-learning during Covid-19: Indonesian sport science education context. *Heliyon* 6:e05410.. doi: 10.1016/j.heliyon.2020.e05410

Taghizadeh, S. K., Rahman, S. A., Nikbin, D., Alam, M. M. D., Alexa, L., Ling Suan, C., et al. (2021). Factors influencing students' continuance usage intention with online learning during the pandemic: a cross-country analysis. *Behav. Inform. Technol.* 41, 1–20. doi: 10.1080/0144929X.2021.1912181

Tajeri Moghadam, M., Zobeidi, T., Sieber, S., and Löhr, K. (2022). Investigating the adoption of precautionary behaviors among young rural adults in South Iran during COVID-19. *Front. Public Health* 10:787929. doi: 10.3389/fpubh.2022.787929

Tao, D., Fu, P., Wang, Y., Zhang, T., and Qu, X. (2019). Key characteristics in designing massive open online courses (MOOCs) for user acceptance: an application of the extended technology acceptance model. *Interact. Learn. Environ.* 30, 1–14. doi: 10.1080/10494820.2019.1695214

Tarhini, A., Hone, K., and Liu, X. (2014). The effects of individual differences on e-learning users' behaviour in developing countries: a structural equation model. *Comput. Hum. Behav.* 41, 153–163. doi: 10.1016/j.chb.2014.09.020

Tarhini, A., Hone, K., Liu, X., and Tarhini, T. (2017). Examining the moderating effect of individual-level cultural values on users' acceptance of E-learning in developing countries: a structural equation modeling of an extended technology acceptance model. *Interact. Learn. Environ.* 25, 306–328. doi: 10.1080/10494820.2015.1122635

Teo, T., Sang, G., Mei, B., and Hoi, C. K. W. (2018). Investigating pre-service teachers' acceptance of web 2.0 technologies in their future teaching: a Chinese perspective. *Interact. Learn. Environ.* 27, 530–546. doi: 10.1080/10494820.2018.1489290

Teo, T., Zhou, M., Fan, A. C. W., and Huang, F. (2019). Factors that influence university students' intention to use Moodle: a study in Macau. *Educ. Technol. Res. Dev.* 67, 749–766. doi: 10.1007/s11423-019-09650-x

Tsai, T. H., Chang, H. T., and Ho, Y. L. (2016). Perceptions of a specific family communication application among grandparents and grandchildren: an extension of the technology acceptance model. *PLoS One* 11:e0156680. doi: 10.1371/journal.pone.0156680

Tsai, C. W., Tsai, S. H., Chen, Y. Y., and Lee, W. L. (2014). A study of nursing competency, career self-efficacy and professional commitment among nurses in Taiwan. *Contemp. Nurse* 49, 96–102. doi: 10.1080/10376178.2014.11081959

Valizadeh, N., Esfandiyari Bayat, S., Bijani, M., Hayati, D., Viira, A. H., Tanaskovik, V., et al. (2021). Understanding farmers' intention towards the management and conservation of wetlands. *Land* 10:860. doi: 10.3390/land10080860

Venkatesh, V., and Bala, H. (2008). Technology acceptance model 3 and a research agenda on interventions. *Decis. Sci.* 39, 273–315. doi: 10.1111/j.1540-5915.2008.00192.x

Venkatesh, V., and Davis, F. D. (2000). A theoretical extension of the technology acceptance model: four longitudinal field studies. *Manag. Sci.* 46, 186–204. doi: 10.1287/mnsc.46.2.186.11926

Venkatesh, V., Morris, M. G., Davis, G. B., and Davis, F. D. (2003). User acceptance of information technology: toward a unified view. *MIS Q.* 27, 425–478. doi: 10.2307/30036540

Vijayasarathy, L. R. (2004). Predicting consumer intentions to use on-line shopping: the case for an augmented technology acceptance model. *Inf. Manag.* 41, 747–762. doi: 10.1016/j.im.2003.08.011

World Health Organisation (2021). WHO Coronavirus (COVID-19) Dashboard. https://covid19.who.int/

Wu, B., and Zhang, C. (2014). Empirical study on continuance intentions towards E-learning 2.0 systems. *Behav. Inform. Technol.* 33, 1027–1038. doi: 10.1080/0144929X.2014.934291

Yaghoubi Farani, A., Karimi, S., Izadi, N., and Ataei, P. (2019). Effect of virtual social networks on entrepreneurial behaviour of agriculture students in Iran. *Appl. Econ.* 51, 2326–2335. doi: 10.1080/00036846.2018.1543940

Yazdanpanah, M., Abadi, B., Komendantova, N., Zobeidi, T., and Sieber, S. (2020). Some at risk for COVID-19 are reluctant to take precautions, but others are not: a case from rural in southern Iran. *Front. Public Health* 8:562300. doi: 10.3389/fpubh.2020.562300

Yazdanpanah, M., Zobeidi, T., Moghadam, M. T., Komendantova, N., Loehr, K., and Sieber, S. (2021). Cognitive theory of stress and farmers' responses to the COVID 19 shock; a model to assess coping behaviors with stress among farmers in southern Iran. *Int. J. Disaster Risk Reduct*. 64:102513. doi: 10.1016/j.ijdrr.2021.102513

Yuen, A. H., and Ma, W. W. (2008). Exploring teacher acceptance of e-learning technology. Asia Pac. J. Teach. Educ. 36, 229–243. doi: 10.1080/13598660802232779

Zamani-Miandashti, N., and Ataei, P. (2015). College students' perceptions of short message service-supported collaborative learning. *Innov. Educ. Teach. Int.* 52, 426–436. doi: 10.1080/14703297.2014.900453

Zhang, Y. (2005). Age, gender, and internet attitudes among employees in the business world. *Comput. Hum. Behav.* 21, 1–10. doi: 10.1016/j.chb.2004.02.006

Zhao, L., Lu, Y., Wang, B., and Huang, W. (2011). What makes them happy and curious online? An empirical study on high school students' internet use from a self-determination theory perspective. *Comput. Educ.* 56, 346–356. doi: 10.1016/j.compedu.2010.08.006

Zobeidi, T., Yaghoubi, J., and Yazdanpanah, M. (2022). Exploring the motivational roots of farmers' adaptation to climate change-induced water stress through incentives or norms. *Sci. Rep.* 12, 1–10. doi: 10.1038/s41598-022-19384-1

Zobeidi, T., Yazdanpanah, M., Komendantova, N., Sieber, S., and Löhr, K. (2021). Factors affecting smallholder farmers' technical and non-technical adaptation responses to drought in Iran. *J. Environ. Manag.* 298:113552. doi: 10.1016/j.jenvman.2021.113552